Accelerator Status Report

Run Schedule: Sept. 27, 2010 to December 23, 2010
Oct. 6, 2010
Injector    Energy 61.76 MeV
· The injector orbits are not reproducible sometimes when the GUN HV is turned on after turning off. As of now we do not know why it behaves like this. It was suggested that we use the injector steering script to recover the orbits. EGG and injector on-calls will support this.

· (10/4) Reza adjusted the injector. When Hall C is ready for beam Reza will check the injector again.

· The Wien was set to 29 degrees.

· Gun#2 operates at 130 kV. The gun HV is not stable sometimes, which would trip off. The EGG is aware of this. Inform the EGG of this whenever it happens.
· Injector needs a revisit to make three beams coincide.
Accelerator   Linac Energy 549.00 MeV
· (10/6) Shunt adder MAN5S03 tripped off while cycling. The control module was replaced.

· (10/6) Arc9A tripped off due to cooling water problems. The power supply needs to be flushed.

· (10/5) It was found that the accelerator energy was set a little bit too high. dp/p was about E-3, which should be E-4 or better. Arc 1 dipole was 0.45% too low. All other arc dipoles were very close to the design. LEM fudge factor (1.032) in SL was too high. Yesterday’s day shift was utilized to correct the problem. Arc 1 was put on design. The SL fudge factor was lowered and it is now 1.007.
· CHLT4 was contaminated. The work was done yesterday to get rid of the contamination. We should be able to restore beam this afternoon (8/29). 

· Pass 5 CW beam (6uA) was established to the BSY dump. The machine optics is reasonable although the FOPT data shows that the optics may need some improvement. The apertures are good.
· Hall B took (pass 5, 20nA) beam for about three shifts. The beam was good.

· (9/27) The SAD ended.

Hall A---Pass 5, DVCS
· (9/29) Hall A started to take beam (low current)

· Planning to take beam tonight (through the Compton).
Hall B ---Pass 5, Primex II
· Hall B beam was not stable sometimes.
· Hall B took beam for three shifts. The hall reported that its beam was fine.
Hall C--- Pass 1, QWeak 
· Not ready for beam yet.

· Hall C Compton vacuum valves have a separate FSD node.

· Not ready for beam until October 5.

